Author's response to reviews

Title: Feedback GAP: study protocol for a cluster-randomized trial of goal setting and action plans to increase the effectiveness of audit and feedback interventions in primary care

Authors:

Noah M Ivers (noahivers@gmail.com)
Karen Tu (karen.tu@ices.on.ca)
Jill Francis (j.francis@abdn.ac.uk)
Jan Barnsley (jan.barnsley@utoronto.ca)
Baiju Shah (baiju.shah@ices.on.ca)
Ross Upshur (ross.upshur@sunnybrook.ca)
Alex Kiss (alex.kiss@ices.on.ca)
Jeremy M Grimshaw (jgrimshaw@ohri.ca)
Merrick Zwarenstein (merrick.zwarenstein@ices.on.ca)

Version: 6 Date: 14 December 2010

Author's response to reviews:

Dear Copy-Editors

We are grateful for your advice and have incorporated all the recommended changes.

Sincerely,

Noah Ivers
on behalf of the Feedback GAP team.